Title: Role of the VEGF ligand to receptor ratio in the progression of mismatch repair-proficient colorectal cancer

Authors:

Manuela Eppenberger (meliepp@hotmail.com)
Inti Zlobec (izlobec@uhbs.ch)
Daniel Baumhoer (dbaumhoer@mac.com)
Luigi Terracciano (lterracciano@uhbs.ch)
Alessandro Lugli (alugli@uhbs.ch)

Version: 7 Date: 9 March 2010

Author’s response to reviews:

Only two typographical errors were corrected, namely (1) on page 5 a "-" was added between "repair and proficient", paragraph 1 and (2) on page 9, third paragraph, 4th line, should be "of" instead of "if".